Author's response to reviews

Title: Consumer Knowledge and Availability Of Maternal And Child Health Services: A Challenge For Achieving MDG 4 And 5 In Southeast Nigeria

Authors:

Nwala EK Emmanuel Mr (emmanuel.nwala@yahoo.com)
Ebunoha GN Gladys Mrs (gladysebunoha@yahoo.com)
Agusiegbe GO Gladys Mrs (goagusiegbe@unn.edu.ng)
Ugwu CU Cosmas Mr (uchecos@yahoo.com)

Version: 2 Date: 7 February 2012

Author's response to reviews:

This work evaluated the consumer knowledge and availability of maternal and child health services in rural Nigeria, and reflected on how the knowledge of available health services challenges the attainment of MGDs 4 & 5. There is growing interest in the attainment of this goal which has challenged numerous governmental and non-governmental health organizations to provide relief packages. However, consumers can use services they know is available and consumer knowledge of available services can invariably influence provider practices. Therefore, publishing this paper in this journal will increase readership for both consumers of health services, providers, as well well as informing future policy decisions.

However, this paper was previously submitted to Israel Journal of Health Policy but was cross transferred because it was out of scope as the journal focuses on health in Israel.

Many thanks for your anticipated cooperation